Author’s response to reviews

Title: The reliability of toe systolic pressure and the toe brachial index in patients with diabetes

Authors:

Mary T Romanos (mary_romanos@hotmail.com)
Anita Rasovic (a.raspovic@latrobe.edu.au)
Byron M Perrin (b.perrin@latrobe.edu.au)

Version: 4 Date: 11 December 2010

Author’s response to reviews:

The major change has been made to the manuscript as outlined in your email dated 7/12/10.

The minor changes have not been made as there was difficulty cropping the figures. It would be appreciative if you could make these changes.

Kind Regards,

Mary Romanos